\begin{thebibliography}{}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\providecommand{\url}[1]{\href{#1}{#1}}
\providecommand{\dodoi}[1]{doi:~\href{http://doi.org/#1}{\nolinkurl{#1}}}
\providecommand{\doeprint}[1]{\href{http://ascl.net/#1}{\nolinkurl{http://ascl.net/#1}}}
\providecommand{\doarXiv}[1]{\href{https://arxiv.org/abs/#1}{\nolinkurl{https://arxiv.org/abs/#1}}}

\bibitem[{{Alcaino}(1975)}]{Alcaino1975}
{Alcaino}, G. 1975, \aaps, 21, 15

\bibitem[{{Anderson} {et~al.}(2008){Anderson}, {Sarajedini}, {Bedin}, {King},
  {Piotto}, {Reid}, {Siegel}, {Majewski}, {Paust}, {Aparicio}, {Milone},
  {Chaboyer}, \& {Rosenberg}}]{Anderson2008}
{Anderson}, J., {Sarajedini}, A., {Bedin}, L.~R., {et~al.} 2008, \aj, 135,
  2055, \dodoi{10.1088/0004-6256/135/6/2055}

\bibitem[{Barber {et~al.}(1996)Barber, Dobkin, \& Huhdanpaa}]{Barber1996}
Barber, C.~B., Dobkin, D.~P., \& Huhdanpaa, H. 1996, ACM Transactions on
  Mathematical Software (TOMS), 22, 469

\bibitem[{{Bastian} \& {Lardo}(2015)}]{Bastian2015}
{Bastian}, N., \& {Lardo}, C. 2015, \mnras, 453, 357,
  \dodoi{10.1093/mnras/stv1661}

\bibitem[{Bastian \& Lardo(2018)}]{Bastian2018}
Bastian, N., \& Lardo, C. 2018, Annual Review of Astronomy and Astrophysics,
  56, 83

\bibitem[{Bekki \& Chiba(2002)}]{Bekki2002}
Bekki, K., \& Chiba, M. 2002, The Astrophysical Journal, 566, 245,
  \dodoi{10.1086/337984}

\bibitem[{{Boudreaux} \& {Chaboyer}(2023)}]{Boudreaux2023}
{Boudreaux}, E.~M., \& {Chaboyer}, B.~C. 2023, \apj, 944, 129,
  \dodoi{10.3847/1538-4357/acb685}

\bibitem[{{Boylan-Kolchin}(2018)}]{Boy18}
{Boylan-Kolchin}, M. 2018, \mnras, 1423, \dodoi{10.1093/mnras/sty1490}

\bibitem[{Brodie \& Strader(2006)}]{Bro06}
Brodie, J.~P., \& Strader, J. 2006, Annu. Rev. Astron. Astrophys., 44, 193

\bibitem[{{Brown} {et~al.}(1991){Brown}, {Burkert}, \& {Truran}}]{Brown1991}
{Brown}, J.~H., {Burkert}, A., \& {Truran}, J.~W. 1991, \apj, 376, 115,
  \dodoi{10.1086/170260}

\bibitem[{{Brown} {et~al.}(1995){Brown}, {Burkert}, \& {Truran}}]{Brown1995}
---. 1995, \apj, 440, 666, \dodoi{10.1086/175304}

\bibitem[{{Cadelano} {et~al.}(2022){Cadelano}, {Dalessandro}, {Salaris},
  {Bastian}, {Mucciarelli}, {Saracino}, {Martocchia}, \&
  {Cabrera-Ziri}}]{Cadelano2022}
{Cadelano}, M., {Dalessandro}, E., {Salaris}, M., {et~al.} 2022, \apjl, 924,
  L2, \dodoi{10.3847/2041-8213/ac424a}

\bibitem[{{Carretta}(2006)}]{Carretta2006}
{Carretta}, E. 2006, \aj, 131, 1766, \dodoi{10.1086/499565}

\bibitem[{{Carretta}(2013)}]{Carretta2013}
---. 2013, \aap, 557, A128, \dodoi{10.1051/0004-6361/201322103}

\bibitem[{{Carretta}(2015)}]{Carretta2015}
---. 2015, \apj, 810, 148, \dodoi{10.1088/0004-637X/810/2/148}

\bibitem[{{Carretta} {et~al.}(2004){Carretta}, {Bragaglia}, \&
  {Cacciari}}]{Carretta2004}
{Carretta}, E., {Bragaglia}, A., \& {Cacciari}, C. 2004, \apjl, 610, L25,
  \dodoi{10.1086/423034}

\bibitem[{Carretta {et~al.}(2010)Carretta, Bragaglia, Gratton, Recio-Blanco,
  Lucatello, D'Orazi, \& Cassisi}]{Carretta2010}
Carretta, E., Bragaglia, A., Gratton, R.~G., {et~al.} 2010, Astronomy \&
  Astrophysics, 516, A55

\bibitem[{{Colgan} {et~al.}(2016){Colgan}, {Kilcrease}, {Magee}, { Sherrill},
  {Abdallah}, {Hakel}, {Fontes}, {Guzik}, \& {Mussack}}]{Colgan2016}
{Colgan}, J., {Kilcrease}, D.~P., {Magee}, N.~H., {et~al.} 2016, in APS Meeting
  Abstracts, Vol. 2016, APS Division of Atomic, Molecular and Optical Physics
  Meeting Abstracts, D1.008

\bibitem[{{de Mink} {et~al.}(2009){de Mink}, {Pols}, {Langer}, \&
  {Izzard}}]{deMink2009}
{de Mink}, S.~E., {Pols}, O.~R., {Langer}, N., \& {Izzard}, R.~G. 2009, \aap,
  507, L1, \dodoi{10.1051/0004-6361/200913205}

\bibitem[{{Decressin} {et~al.}(2007){Decressin}, {Meynet}, {Charbonnel},
  {Prantzos}, \& {Ekstr{\"o}m}}]{Decressin2007}
{Decressin}, T., {Meynet}, G., {Charbonnel}, C., {Prantzos}, N., \&
  {Ekstr{\"o}m}, S. 2007, \aap, 464, 1029, \dodoi{10.1051/0004-6361:20066013}

\bibitem[{{Denissenkov} \& {Hartwick}(2014)}]{Denissenkov2014}
{Denissenkov}, P.~A., \& {Hartwick}, F.~D.~A. 2014, \mnras, 437, L21,
  \dodoi{10.1093/mnrasl/slt133}

\bibitem[{{D'Ercole} {et~al.}(2010){D'Ercole}, {D'Antona}, {Ventura},
  {Vesperini}, \& {McMillan}}]{DErcole2010}
{D'Ercole}, A., {D'Antona}, F., {Ventura}, P., {Vesperini}, E., \& {McMillan},
  S. L.~W. 2010, \mnras, 407, 854, \dodoi{10.1111/j.1365-2966.2010.16996.x}

\bibitem[{{D'Ercole} {et~al.}(2008){D'Ercole}, {Vesperini}, {D'Antona},
  {McMillan}, \& {Recchi}}]{DErcole2008}
{D'Ercole}, A., {Vesperini}, E., {D'Antona}, F., {McMillan}, S. L.~W., \&
  {Recchi}, S. 2008, \mnras, 391, 825, \dodoi{10.1111/j.1365-2966.2008.13915.x}

\bibitem[{{Dotter}(2016)}]{Dotter2016}
{Dotter}, A. 2016, \apjs, 222, 8, \dodoi{10.3847/0067-0049/222/1/8}

\bibitem[{Dotter {et~al.}(2008)Dotter, Chaboyer, Jevremovi{\'c}, Kostov, Baron,
  \& Ferguson}]{Dotter2008}
Dotter, A., Chaboyer, B., Jevremovi{\'c}, D., {et~al.} 2008, The Astrophysical
  Journal Supplement Series, 178, 89

\bibitem[{{Dotter} {et~al.}(2015){Dotter}, {Ferguson}, {Conroy}, { Milone},
  {Marino}, \& {Yong}}]{Dotter2015}
{Dotter}, A., {Ferguson}, J.~W., {Conroy}, C., {et~al.} 2015, \mnras, 446,
  1641, \dodoi{10.1093/mnras/stu2170}

\bibitem[{Ferguson(1973)}]{Ferguson1973}
Ferguson, T.~S. 1973, The annals of statistics, 209

\bibitem[{{Gratton} {et~al.}(2004){Gratton}, {Sneden}, \&
  {Carretta}}]{Gratton2004}
{Gratton}, R., {Sneden}, C., \& {Carretta}, E. 2004, \araa, 42, 385,
  \dodoi{10.1146/annurev.astro.42.053102.133945}

\bibitem[{{Gratton} {et~al.}(2012){Gratton}, {Carretta}, \&
  {Bragaglia}}]{Gratton2012}
{Gratton}, R.~G., {Carretta}, E., \& {Bragaglia}, A. 2012, Astronomy and
  Astrophysics Reviews, 20, 50, \dodoi{10.1007/s00159-012-0050-3}

\bibitem[{{Gratton} {et~al.}(2011){Gratton}, {Lucatello}, {Carretta},
  {Bragaglia}, {D'Orazi}, \& {Momany}}]{Gratton2011}
{Gratton}, R.~G., {Lucatello}, S., {Carretta}, E., {et~al.} 2011, \aap, 534,
  A123, \dodoi{10.1051/0004-6361/201117690}

\bibitem[{{Grevesse} {et~al.}(2007){Grevesse}, {Asplund}, \&
  {Sauval}}]{Grevesse2007}
{Grevesse}, N., {Asplund}, M., \& {Sauval}, A.~J. 2007, \ssr, 130, 105,
  \dodoi{10.1007/s11214-007-9173-7}

\bibitem[{{Gustafsson} {et~al.}(2008){Gustafsson}, {Edvardsson}, {Eriksson},
  {J{\o} rgensen}, {Nordlund}, \& {Plez}}]{Gustafsson2008}
{Gustafsson}, B., {Edvardsson}, B., {Eriksson}, K., {et~al.} 2008, \aap, 486,
  951, \dodoi{10.1051/0004-6361:200809724}

\bibitem[{{Hong} {et~al.}(2021){Hong}, {Lim}, {Chung}, {Kim}, {Han}, \&
  {Lee}}]{Hong2021}
{Hong}, S., {Lim}, D., {Chung}, C., {et~al.} 2021, \aj, 162, 130,
  \dodoi{10.3847/1538-3881/ac0ce6}

\bibitem[{{Hudson} \& {Robison}(2018)}]{Hud18}
{Hudson}, M.~J., \& {Robison}, B. 2018, Monthly Notices of the Royal
  Astronomical Society, 477, 3869, \dodoi{10.1093/mnras/sty844}

\bibitem[{{Husser} {et~al.}(2013){Husser}, {Wende-von Berg}, {Dreizler}, {
  Homeier}, {Reiners}, {Barman}, \& {Hauschildt}}]{Husser2013}
{Husser}, T.~O., {Wende-von Berg}, S., {Dreizler}, S., {et~al.} 2013, \aap,
  553, A6, \dodoi{10.1051/0004-6361/201219058}

\bibitem[{{Jord{\'a}n} {et~al.}(2002){Jord{\'a}n}, {C{\^o}t{\'e}}, {West}, \&
  {Marzke}}]{Jordan2002}
{Jord{\'a}n}, A., {C{\^o}t{\'e}}, P., {West}, M.~J., \& {Marzke}, R.~O. 2002,
  \apjl, 576, L113, \dodoi{10.1086/343759}

\bibitem[{{Kalirai} \& {Richer}(2010)}]{Kalirai2010}
{Kalirai}, J.~S., \& {Richer}, H.~B. 2010, Philosophical Transactions of the
  Royal Society of London Series A, 368, 755, \dodoi{10.1098/rsta.2009.0257}

\bibitem[{Kostogryz {et~al.}(2023)Kostogryz, Shapiro, Witzke, Grant, Wakeford,
  Stevenson, Solanki, \& Gizon}]{Kostogryz2023}
Kostogryz, N., Shapiro, A.~I., Witzke, V., {et~al.} 2023, Research Notes of the
  AAS, 7, 39, \dodoi{10.3847/2515-5172/acc180}

\bibitem[{Kravtsov \& Gnedin(2005)}]{Kra05}
Kravtsov, A.~V., \& Gnedin, O.~Y. 2005, The Astrophysical Journal, 623, 650

\bibitem[{Kurucz(1993)}]{Kurucz1993}
Kurucz, R.-L. 1993, Kurucz CD-Rom, 13

\bibitem[{{Latour} {et~al.}(2019){Latour}, {Husser}, {Giesers}, {Kamann},
  {G{\"o}ttgens}, {Dreizler}, {Brinchmann}, { Bastian}, {Wendt}, {Weilbacher},
  \& {Molinski}}]{Latour2019}
{Latour}, M., {Husser}, T.~O., {Giesers}, B., {et~al.} 2019, \aap, 631, A14,
  \dodoi{10.1051/0004-6361/201936242}

\bibitem[{{Marigo} \& {Aringer}(2009)}]{Marigo2009}
{Marigo}, P., \& {Aringer}, B. 2009, \aap, 508, 1539,
  \dodoi{10.1051/0004-6361/200912598}

\bibitem[{{Marigo} {et~al.}(2022){Marigo}, {Aringer}, {Girardi}, \& {
  Bressan}}]{Marigo2022}
{Marigo}, P., {Aringer}, B., {Girardi}, L., \& { Bressan}, A. 2022, \apj, 940,
  129, \dodoi{10.3847/1538-4357/ac9b40}

\bibitem[{Marino {et~al.}(2015)Marino, Milone, Karakas, Casagrande, Yong,
  Shingles, Da~Costa, Norris, Stetson, Lind, Asplund, Collet, Jerjen, Sbordone,
  Aparicio, \& Cassisi}]{Marino2015}
Marino, A.~F., Milone, A.~P., Karakas, A.~I., {et~al.} 2015, Monthly Notices of
  the Royal Astronomical Society, 450, 815, \dodoi{10.1093/mnras/stv420}

\bibitem[{Martocchia {et~al.}(2019)Martocchia, Dalessandro, Lardo,
  Cabrera-Ziri, Bastian, Kozhurina-Platais, Salaris, Chantereau, Geisler,
  Hilker, Kacharov, Larsen, Mucciarelli, Niederhofer, Platais, \&
  Usher}]{Martocchia2019}
Martocchia, S., Dalessandro, E., Lardo, C., {et~al.} 2019, Monthly Notices of
  the Royal Astronomical Society, 487, 5324, \dodoi{10.1093/mnras/stz1596}

\bibitem[{{Milone} {et~al.}(2012){Milone}, {Piotto}, {Bedin}, {King},
  {Anderson}, {Marino}, {Bellini}, { Gratton}, {Renzini}, {Stetson}, {Cassisi},
  {Aparicio}, {Bragaglia}, {Carretta}, { D'Antona}, {Di Criscienzo},
  {Lucatello}, { Monelli}, \& {Pietrinferni}}]{Milone2012}
{Milone}, A.~P., {Piotto}, G., {Bedin}, L.~R., {et~al.} 2012, \apj, 744, 58,
  \dodoi{10.1088/0004-637X/744/1/58}

\bibitem[{{Milone} {et~al.}(2015{\natexlab{a}}){Milone}, {Marino}, {Piotto},
  {Renzini}, {Bedin}, {Anderson}, {Cassisi}, { D'Antona}, {Bellini}, {Jerjen},
  {Pietrinferni}, \& {Ventura}}]{Milone2015}
{Milone}, A.~P., {Marino}, A.~F., {Piotto}, G., {et~al.} 2015{\natexlab{a}},
  \apj, 808, 51, \dodoi{10.1088/0004-637X/808/1/51}

\bibitem[{{Milone} {et~al.}(2015{\natexlab{b}}){Milone}, {Marino}, {Piotto},
  {Bedin}, {Anderson}, {Renzini}, {King}, {Bellini}, {Brown}, {Cassisi},
  {D'Antona}, {Jerjen}, {Nardiello}, {Salaris}, {Marel}, {Vesperini}, {Yong},
  {Aparicio}, {Sarajedini}, \& {Zoccali}}]{Milone2015b}
---. 2015{\natexlab{b}}, \mnras, 447, 927, \dodoi{10.1093/mnras/stu2446}

\bibitem[{{Milone} {et~al.}(2017){Milone}, {Piotto}, {Renzini}, {Marino},
  {Bedin}, {Vesperini}, {D'Antona}, {Nardiello}, {Anderson}, {King}, {Yong},
  {Bellini}, {Aparicio}, {Barbuy}, {Brown}, {Cassisi}, {Ortolani}, {Salaris},
  {Sarajedini}, \& {van der Marel}}]{Milone2017}
{Milone}, A.~P., {Piotto}, G., {Renzini}, A., {et~al.} 2017, \mnras, 464, 3636,
  \dodoi{10.1093/mnras/stw2531}

\bibitem[{{Milone} {et~al.}(2018){Milone}, {Marino}, {Renzini}, {D'Antona},
  {Anderson}, {Barbuy}, {Bedin}, {Bellini}, {Brown}, {Cassisi}, {Cordoni},
  {Lagioia}, {Nardiello}, {Ortolani}, {Piotto}, {Sarajedini}, {Tailo}, {van der
  Marel}, \& {Vesperini}}]{Milone2018}
{Milone}, A.~P., {Marino}, A.~F., {Renzini}, A., {et~al.} 2018, \mnras, 481,
  5098, \dodoi{10.1093/mnras/sty2573}

\bibitem[{{Pasquato} \& {Milone}(2019)}]{Pasquato2019}
{Pasquato}, M., \& {Milone}, A. 2019, arXiv e-prints, arXiv:1906.04983,
  \dodoi{10.48550/arXiv.1906.04983}

\bibitem[{Pedregosa {et~al.}(2011)Pedregosa, Varoquaux, Gramfort, Michel,
  Thirion, Grisel, Blondel, Prettenhofer, Weiss, Dubourg, Vanderplas, Passos,
  Cournapeau, Brucher, Perrot, \& Duchesnay}]{scikit-learn}
Pedregosa, F., Varoquaux, G., Gramfort, A., {et~al.} 2011, Journal of Machine
  Learning Research, 12, 2825

\bibitem[{{Peebles} \& {Dicke}(1968)}]{Peebles1968}
{Peebles}, P.~J.~E., \& {Dicke}, R.~H. 1968, \apj, 154, 891,
  \dodoi{10.1086/149811}

\bibitem[{Peng {et~al.}(2011)Peng, Ferguson, Goudfrooij, Hammer, Lucey, Marzke,
  Puzia, Carter, Balcells, Bridges, {et~al.}}]{Pen11}
Peng, E.~W., Ferguson, H.~C., Goudfrooij, P., {et~al.} 2011, The Astrophysical
  Journal, 730, 23

\bibitem[{Piotto(2018)}]{HUGS}
Piotto, G. 2018, HST UV Globular Cluster Survey ("HUGS"),  STScI/MAST,
  \dodoi{10.17909/T9810F}

\bibitem[{{Piotto} {et~al.}(2007){Piotto}, {Bedin}, {Anderson}, {King},
  {Cassisi}, {Milone}, {Villanova}, { Pietrinferni}, \& {Renzini}}]{Piotto2007}
{Piotto}, G., {Bedin}, L.~R., {Anderson}, J., {et~al.} 2007, The Astrophysical
  Journal Letters, 661, L53, \dodoi{10.1086/518503}

\bibitem[{{Piotto} {et~al.}(2015){Piotto}, {Milone}, {Bedin}, {Anderson},
  {King}, {Marino}, {Nardiello}, { Aparicio}, {Barbuy}, {Bellini}, {Brown},
  {Cassisi}, {Cool}, {Cunial}, {Dalessandro }, {D'Antona}, {Ferraro},
  {Hidalgo}, { Lanzoni}, {Monelli}, {Ortolani}, {Renzini}, {Salaris},
  {Sarajedini}, {van der Marel}, {Vesperini}, \& {Zoccali}}]{Piotto2015}
{Piotto}, G., {Milone}, A.~P., {Bedin}, L.~R., {et~al.} 2015, \aj, 149, 91,
  \dodoi{10.1088/0004-6256/149/3/91}

\bibitem[{{Plez}(2008)}]{Plez2008}
{Plez}, B. 2008, Physica Scripta Volume T, 133, 014003,
  \dodoi{10.1088/0031-8949/2008/T133/014003}

\bibitem[{{Prantzos} {et~al.}(2007){Prantzos}, {Charbonnel}, \&
  {Iliadis}}]{Prantzos2007}
{Prantzos}, N., {Charbonnel}, C., \& {Iliadis}, C. 2007, \aap, 470, 179,
  \dodoi{10.1051/0004-6361:20077205}

\bibitem[{Renzini(2008)}]{Renzini2008}
Renzini, A. 2008, Monthly Notices of the Royal Astronomical Society, 391, 354,
  \dodoi{10.1111/j.1365-2966.2008.13892.x}

\bibitem[{{Richer} {et~al.}(1991){Richer}, {Fahlman}, {Buonanno}, {Fusi Pecci},
  {Searle}, \& {Thompson}}]{Richer1991}
{Richer}, H.~B., {Fahlman}, G.~G., {Buonanno}, R., {et~al.} 1991, \apj, 381,
  147, \dodoi{10.1086/170637}

\bibitem[{Rousseeuw(1987)}]{ROUSSEEUW198753}
Rousseeuw, P.~J. 1987, Journal of Computational and Applied Mathematics, 20,
  53, \dodoi{https://doi.org/10.1016/0377-0427(87)90125-7}

\bibitem[{Salaris \& Cassisi(2005)}]{Salaris2005}
Salaris, M., \& Cassisi, S. 2005, Evolution of stars and stellar populations
  (John Wiley \& Sons)

\bibitem[{{Sandage}(1953)}]{Sandage1953}
{Sandage}, A.~R. 1953, \aj, 58, 61, \dodoi{10.1086/106822}

\bibitem[{Shahapure \& Nicholas(2020)}]{Shahapure2020}
Shahapure, K.~R., \& Nicholas, C. 2020, in 2020 IEEE 7th International
  Conference on Data Science and Advanced Analytics (DSAA), 747--748,
  \dodoi{10.1109/DSAA49011.2020.00096}

\bibitem[{Smith(1987)}]{Smith1987}
Smith, G.~H. 1987, Publications of the Astronomical Society of the Pacific, 99,
  67, \dodoi{10.1086/131958}

\bibitem[{Sneden {et~al.}(1992)Sneden, Kraft, Prosser, \& Langer}]{Sneden1992}
Sneden, C., Kraft, R.~P., Prosser, C.~F., \& Langer, G. 1992, The Astronomical
  Journal, 104, 2121

\bibitem[{{Suntzeff} \& {Kraft}(1996)}]{Suntzeff1996}
{Suntzeff}, N.~B., \& {Kraft}, R.~P. 1996, \aj, 111, 1913,
  \dodoi{10.1086/117930}

\bibitem[{{Valle} {et~al.}(2022){Valle}, {Dell'Omodarme}, \&
  {Tognelli}}]{Valle2022}
{Valle}, G., {Dell'Omodarme}, M., \& {Tognelli}, E. 2022, \aap, 658, A141,
  \dodoi{10.1051/0004-6361/202142454}

\bibitem[{van~den Bergh(2010)}]{Vanderburg2010}
van~den Bergh, S. 2010, The Astronomical Journal, 140, 1043,
  \dodoi{10.1088/0004-6256/140/4/1043}

\bibitem[{{Ventura} \& {D'Antona}(2009)}]{Ventura2009}
{Ventura}, P., \& {D'Antona}, F. 2009, \aap, 499, 835,
  \dodoi{10.1051/0004-6361/200811139}

\bibitem[{{Ventura} {et~al.}(2001){Ventura}, {D'Antona}, {Mazzitelli}, \&
  {Gratton}}]{Ventura2001}
{Ventura}, P., {D'Antona}, F., {Mazzitelli}, I., \& {Gratton}, R. 2001, \apjl,
  550, L65, \dodoi{10.1086/319496}

\end{thebibliography}

\section{Introduction}\label{sec:Intro}
Globular clusters (GCs) are among the oldest observable objects in the universe
\citep{Pen11}. They are characterized by high densities with typical half-light
radii of $\le$10 pc \citep{Vanderburg2010}, and typical masses ranging from
$10^{4}$--$10^{5}$ M$_{\odot}$ \citep{Bro06} --- though some GCs are
significantly larger than these typical values \citep[e.g. $\omega$ Cen,
][]{Richer1991}. GCs provide a unique way to probe stellar evolution
\citep{Bau03}, galaxy formation models \citep{Boy18,Kra05}, and dark matter
halo structure \citep{Hud18}.

The traditional view of Globular Clusters was that they consisted of a single
stellar population (SSP, in some publications this is referred to as a Simple
Stellar Population). This view was supported by spectroscopically uniform heavy
element abundances \citep{Carretta2010, Bastian2018} across most clusters (M54
and $\omega$Cen are notable exceptions, see \citet{Marino2015} for further
details), and the lack of evidence for multiple stellar populations (MPs) in
past color-magnitude diagrams of GCs \citep[i.e.][]{Sandage1953, Alcaino1975}.
However, over the last 40 years non-trivial star-to-star light-element
abundance variations have been observed \citep[i.e.][]{Smith1987} and, in the
last two decades, it has been definitively shown that most if not all Milky Way
GCs have MPs \citep{Gratton2004, Gratton2012, Piotto2015}. The lack of
photometric evidence for MPs prior to the 2000, can be attributed to the more
narrow color bands available, until very recently, to ground based photometric
surveys \citep{Milone2017}.

The prevalence of multiple populations in GCs is so distinct that the proposed
definitions for what constitutes a globular cluster now often center the
existence of MPs \citep[e.g.][]{Carretta2010}. Whereas, people have have often tried to categorized objects
as GCs through relations between half-light radius, density, and surface
brightness profile, in fact many objects which are generally thought of as GCs
don't cleanly fit into these cuts \citep{Peebles1968, Brown1991, Brown1995, Bekki2002}.
Consequently, \citet{Carretta2010} proposed a definition of GC based on
observed chemical inhomogeneities in their stellar populations. The modern
understanding of GCs then is not simply one of a dense cluster of stars that
may have chemical inhomogeneities and multiple populations; rather, it is one
where those chemical inhomogeneities and multiple populations themselves are
the defining element of a GC.

All Milky Way globular clusters older than 2 Gyr studied in detail show
populations enriched in He, N, and Na while also being deplete in O and C
\citep{Piotto2015,Bastian2018}. These light element abundance patterns also are
not strongly correlated with variations in heavy element abundance, resulting
in spectroscopically uniform Fe abundances between populations. Further,
high-resolution spectral studies reveal anti-correlations between N-C
abundances, Na-O abundances, and potentially Al-Mg \citep{Sneden1992,
Gratton2012}. Typical stellar fusion reactions can deplete core oxygen;
however, the observed abundances of Na, Al, and Mg cannot be explained by the
CNO cycle \citep{Prantzos2007}. Consequently, globular cluster populations must
be formed by some novel means.

Formation channels for these multiple populations remain a point of debate
among astronomers. Most proposed formation channels consist of some older,
more massive, population of stars polluting the pristine cluster media before a
second population forms, now enriched in heavier elements which they themselves could
not have generated \citep[for a detailed review see ][]{Gratton2012}. The four
primary candidates for these polluters are asymptotic giant branch stars
\citep[AGBs,][]{Ventura2001,DErcole2010}, fast rotating massive stars
\citep[FRMSs,][]{Decressin2007}, super massive stars
\citep[SMSs,][]{Denissenkov2014}, and massive interacting binaries
\citep[MIBs,][]{deMink2009, Bastian2018}. 

Hot hydrogen burning (i.e. proton capture), material transport to the surface, and
material ejection into the intra-cluster media are features of each of these
models and consequently they can all be made to {\it qualitatively} agree with
the observed elemental abundances. However, none of the standard models can
currently account for all specific abundances \citep{Gratton2012}. AGB and FRMS
models are the most promising; however, both models have difficulty reproducing
severe O depletion \citep{Ventura2009,Decressin2007}. Moreover, AGB and FRMS
models require significant mass loss ($\sim 90\%$) between cluster formation
and the current epoch --- implying that a significant fraction of halo stars
formed in GCs \citep{Renzini2008,DErcole2008,Bastian2015}.

In addition to the light-element anti-correlations observed, it is also known
that younger populations are significantly enhanced in Helium
\citep{Piotto2007, Piotto2015, Latour2019}. Depending on the cluster, helium
mass fractions as high as $Y=0.4$ have been inferred \citep[e.g][]{Milone2015}.
However, due to both the relatively high and tight temperature range of partial
ionization for He and the efficiency of gravitational settling in core helium
burning stars, the initial He abundance of globular cluster stars cannot be
observed; consequently, the evidence for enhanced He in GCs originates from
comparison of theoretical stellar isochrones to the observed
color-magnitude-diagrams of globular clusters. Therefore, a careful handling of
chemistry is essential when modeling with the aim of discriminating between
MPs; yet, only a very limited number of GCs have been studied with
chemically self-consistent (structure and atmosphere) isochrones
\citep[e.g.][NGC 6752]{Dotter2015}. 


NGC 2808 is the prototype globular cluster to host Multiple Populations.
Various studies since 2007 have identified that it may host anywhere from 2-5
stellar populations. These populations have been identified both
spectroscopically \citep[i.e.][]{Carretta2004, Carretta2006, Carretta2010, Gratton2011, Carretta2015, Hong2021} and photometrically \citep[i.e.][]{Piotto2007, Piotto2015, Milone2015, Milone2017, Pasquato2019}. Note
that recent work \citep{Valle2022} calls into question the statistical
significance of the detections of more than 2 populations in the spectroscopic
data. Here we present new, chemically self-consistent modeling of the
photometry of the two extreme populations of NGC 2808 identified by
\citet{Milone2015}, populations A and E. We use archival photometry from the
Hubble UV Globular Cluster Survey (HUGS) \citep{Piotto2015, Milone2017} in the
F275W and F814W passbands to characterize multiple populations in NGC 2808
\citep{Milone2015, Milone2015b}. Additionally, we present a likelihood analysis of the
photometric data of NGC 2808 to determine the number of populations present in
the cluster.


